

Preventing Child Maltreatment: Beneficial Side Effects of Public Childcare Provision

Malte Sandner Stephan L. Thomsen Libertad González

October 2020

Barcelona GSE Working Paper Series Working Paper nº 1207

Preventing Child Maltreatment: Beneficial Side Effects of Public Childcare Provision*

Malte Sandner[†] (Institute for Employment Research (IAB) Nuremberg)

Stephan L. Thomsen[‡] (Leibniz Universität Hannover, IZA Bonn, and ZEW Mannheim)

> Libertad González[§] (Universitat Pompeu Fabra and Barcelona GSE)

October 2020

Abstract: We investigate the impact of public childcare provision on the incidence of child maltreatment. For identification, we exploit a government reform that expanded early childcare in Germany, generating large temporal and spatial variation in childcare coverage at the county level. Using high-quality administrative data covering all reported cases of child maltreatment in Germany by county and year, our results show that an increase in childcare slots by one percentage point in a county led to a decline of 1.8% in child maltreatment cases. Our findings suggest that the provision of universal public childcare may be more cost-effective that previously thought.

Keywords: Child maltreatment, child abuse and neglect, early childcare **JEL classification**: J13, J12, I38

^{*}Acknowledgments: The data used in this project are provided by the research data center of the Federal Statistical Office (Statistisches Bundesamt). We thank Gabriela Conti, Thomas Cornelissen, Flavio Cunha, Elisabeth deCao, Christina Felfe, Michael Gilraine, and Adrian Lerche for helpful advice. Our work has further benefited from discussions at the AASLE meetings 2017, the RES conference 2018, the SOLE conference 2018, the SEHO meetings 2018, the CEA annual conference 2018, the ESPE 2018, the annual conference of the German Economic Association 2018, the EALE 2018, the Workshop on Labour and Family Economics (WOLFE) 2018, and the COMPIE 2018. All remaining errors are the responsibility of the authors. This is a revised version of the IZA and HCEO working paper "*The Effects of Universal Public Childcare Provision on Cases of Child Abuse and Neglect*". [†] Malte Sandner (corresponding author), Institute for Employment Research (IAB), Regensburger Str. 100, D-

⁹⁰⁴⁷⁸ Nuremberg, malte.sandner@iab.de.

^{*} Stephan Thomsen, Institute for Economic Policy, Leibniz University Hannover, Königsworther Platz 1, 30167 Hannover, thomsen@wipol.uni-hannover.de.

[§] Libertad González, Dep. of Economics and Business, Universitat Pompeu Fabra, Ramon Trias Fargas, 25-27, 08005 Barcelona, <u>libertad.gonzalez@upf.edu</u>.

1. Introduction

We study the effect of increasing childcare provision on the incidence of child maltreatment. We exploit a reform in Germany that led to substantial increases in childcare coverage in the 2000's, and follow a difference-in-differences approach that leverages the differential rollout of the policy across counties. Our results suggest that increases in childcare coverage lowered (detected) child maltreatment cases in the county. We explore potential mechanisms and provide suggestive evidence that childcare replaced lower-quality informal care and improved mothers' earnings and bargaining power.

Child maltreatment includes all forms of child abuse, including physical abuse, emotional mistreatment, sexual abuse, neglect, negligent treatment, and exploitation of children, all of which result in actual or potential harm to the child's health, survival, development, or dignity. Child maltreatment is a severe problem in many developed countries. For example, the U.S. Department of Health and Human Services estimates approximately 683,000 victims, and the German statistical office reports about 130,000 suspected cases, in 2015 alone (U.S. Department of Health and Human Services 2017; Statistisches Bundesamt, 2016). As child development is a cumulative process (e.g., Cunha et al. 2006; Cunha and Heckman 2007), child maltreatment – particularly at the beginning of life – leads to lifelong suffering in terms of psychosocial and health problems for the affected children.¹

Given the long-term nature of these problems, their associated costs are high. Beyond direct costs, such as those for childhood health care and child welfare, indirect costs of maltreatment may develop from lower employment rates, lower earnings and tax revenues, and increased

¹ Various studies show that child maltreatment has lifelong effects on physical and psychological development and health, as well as on social behavior and life satisfaction (e.g., Ammerman et al. 1986, Hildyard and Wolfe 2002, and Springer et al. 2007). The effects of adverse environments at the start of life are cumulative because of self-productivities, dynamic complementarities, and sensitive periods in skill development (see Heckman and Masso 2014, or Thiel and Thomsen 2013, for a literature review).

crime rates.² For the U.S., Fang et al. (2012) and Wang and Holton (2007) estimate average lifetime costs of \$210,012 (in 2010 dollars) per victim of nonfatal maltreatment, and aggregated costs of more than 100 billion dollars per year. For the UK, Conti et al. (2017) suggest costs of approximately 90,000£ per case of child maltreatment. These numbers illustrate that it is in society's best interest to reduce – and ideally eliminate – the incidence child maltreatment.

Research on preventing the maltreatment of young children indicates that certain targeted and intensive early childhood interventions can be successful. These interventions mostly include intensive at-home consulting for disadvantaged families, with the goal of reducing caregivers' abusive and neglectful parenting behaviors (see summaries of the results in Doyle & Aizer 2018, Levey et al. 2017, or Howard and Brooks-Gunn 2009). However, obtaining access to families at risk and maintaining their participation is challenging for these in-home programs because families may not only feel stigmatized but also have to invest time and effort in participating (Hernandez et al., 2019 for the U.S. and Sandner, 2019 for Germany).

In contrast to these targeted and intensive programs, universal public childcare supports a wide range of families and is therefore less stigmatizing. Although public childcare does not directly focus on reducing child maltreatment, it may influence the risk of child maltreatment because it may change care quality, parental employment, the time that children spend with inadequate caregivers, and the behavior of inadequate caregivers. However, despite the strong relationship between childcare and several domains of families' lives – domains that may also affect adverse parenting –, we are not aware of causal evidence on the extent to which the provision of universal childcare can reduce child maltreatment.

This study is the first to investigate the effects of childcare availability on reported cases of child maltreatment. Our measure for child maltreatment comes from unique high-quality

² E.g., Currie and Tekin (2012) have analyzed the increased incidence of crime due to child maltreatment. For other economic outcomes, see Currie and Spatz-Widom (2010).

administrative data covering all cases of maltreatment in Germany leading to out-of-home placement. For identification, we use a childcare expansion which resulted from a reform in West Germany that included a federal government commitment to provide childcare placements for all children below the age of three.³ While in 2002, childcare places were available for less than 3% of children, by 2015, the reform had initiated a 25% increase in childcare availability in West Germany. We use the plausibly exogenous variation in the speed and level of the expansion across counties and over time, as well as differences in the starting points, in 324 West German counties. Following Berlinski et al. (2009), Havnes and Mogstad (2011), and Bauernschuster et al. (2016), we apply a difference-in-differences approach (DiD), regressing child maltreatment cases on childcare coverage rates in each county, controlling for year and county fixed effects, as well as a set of controls.

Our results show that a one-percentage-point increase in the availability of childcare reduces maltreatment cases for children below age six by approximately 1.8%. This figure suggests that the expansion avoided approximately 20,000 maltreatment cases in our observation period of 2002-2015, compared with a scenario of no childcare expansion. Due to potentially significant unreported cases of child maltreatment, we can interpret our results as lower-bound estimates. As the increase in childcare facilities may have increased the detection of child maltreatment cases (Fitzpatrick et al. 2020, Baron et al. 2020), this higher detection would have led to more – rather than fewer – cases in the official reporting statistics.

We conduct an event-study analysis that allows us to show that counties with different childcare expansion trajectories were on parallel trends in child maltreatment rates before the policy change. Our robustness checks also show that our results are maintained across various

³ The reform was introduced in 2005 in the so-called "child daycare expansion law" (*Tagesbetreuungsausbaugesetz, 2005*), and further specified in 2008 in a second law called "law on support for children" (*Kinderförderungsgesetz, 2008*). The main objectives were to achieve equal opportunity, reduce social disparities, and provide better educational prospects for all children.

alternative specifications and for different subsamples. In addition, we conduct placebo tests by estimating models on older children, not affected by the childcare expansion. These placebo tests allow us to rule out the possibility that changes in the structure or organization of local child protective services (CPS) in response to the childcare expansion may have influenced the incidence of maltreatment.⁴

We believe our results acquired particular relevance in 2020, when many countries have closed childcare centers and schools in order to control the spread of covid-19. Our findings suggest that the closing of childcare centers may exacerbate parental neglect and abuse in vulnerable families, as well as possibly making those cases harder to detect.⁵ The potential costs associated with the increase in child maltreatment should be taken into account in cost-benefit calculations of lockdown measures.

The rich administrative data on child maltreatment give us the opportunity to investigate the mechanisms behind the maltreatment reduction. First, we show that the childcare expansion increased female employment, which likely increased maternal earnings and women's bargaining power in the household. Second, we find that the strongest reduction occurs in households in which a male partner or husband is present, while childcare expansion has no effect in single-mother households. Third, our estimates show the strongest reductions for families who were already in contact with or under the observation of the CPS. Finally, we find that the reduction concentrates on child protection cases that are initiated because of abuse and neglect. Child protection cases that are initiated because of parental overburden are not affected. These findings suggest that the substitution of inadequate informal childcare (by men) in

⁴ In Germany, the child protection service (CPS) is called the youth welfare office (*Jugendamt*). The youth welfare office has similar responsibilities and duties, such as supporting and monitoring families at risk, connecting to support services, and, as a final measure, removing maltreated children from the family, as the CPS does in the U.S. or U.K.

⁵ Baron et al. (2020) show that school closures in Florida in early 2020 led to lower reporting of child maltreatment cases.

families at risk constitutes the main channel for the reduction in child maltreatment cases. This explanation is in line with Lindo et al. (2018), who demonstrate that cases of child maltreatment increase if men, as a main source of informal childcare, spend more time with children.

Our setting is particularly suitable for this analysis for the following three reasons. First, several studies investigating other outcomes of the German childcare expansion, such as child development or fertility, show that the variation in the expansion was independent of county characteristics (e.g., Bauernschuster et al. 2016; Felfe and Lalive 2018). This is supported by our own analysis. Therefore, the daycare expansion for children below age three provides a natural experiment enabling us to identify causal effects. Second, childcare fees are meanstested and depend on available household income. Free childcare is provided for low-income and welfare-receiving families, who are at the highest risk of child maltreatment (e.g., McLoyd, 1990; Paxson and Waldfogel, 2002). Therefore, self-selection due to budgetary constraints is not very likely, particularly for families at risk. Indeed, in this respect, the German childcare provision is similar to the U.S. Department of Health and Human Services 2014). Third, in Germany, the federal government is responsible for child protection legislation. Therefore, no state or county can deviate from it by creating child protection legislation that correlates (intentionally or not) with childcare availability.

Because our paper is the first to analyze the effects of expanding public childcare for young children on child maltreatment, our results answer three major and previously unanswered questions. First, they suggest that not only targeted interventions (e.g., home visiting programs) but also general public policies that substantially change the mode of care can prevent severe cases of maltreatment. In contrast to intensive programs, using universal childcare does not stigmatize families at risk and therefore leads to a much higher take-up rate among these families. The effect of a reduction in child maltreatment thus strengthens support for the policies

of those countries that offer publicly funded universal childcare programs (e.g., France, Germany, the Nordic countries, and the UK). Moreover, the results of this study can influence the discussion in the U.S., where no nationwide universal preschool or early childcare programs are available but where the Obama administration's *Zero to Five* plan, which the Trump administration largely continued, aimed to create similar initiatives.

Second, because maltreatment has strong detrimental effects on children's cognitive and noncognitive development, our findings add to the more general discussion about the channels through which universal public childcare provision affects child development. The findings in the literature on the overall effects of universal public childcare on child development, particularly for children younger than three, are mixed. Baker et al. (2008) and Fort et al. (2019), among others, report negative average effects. In contrast, several recent studies show that public childcare is beneficial for children from families with lower socioeconomic status (e.g., Drange and Havnes 2019, Bitler et al. 2015, Peter et al. 2016, Kottelenberg and Lehrer 2017, Felfe and Lalive 2018).⁶

Because child maltreatment occurs more frequently in more disadvantaged families, our study presents a channel through which childcare positively affects child development therein. Our results indicate that childcare utilization has a positive impact on development, not only through the increased provision of stimulating nurseries or peers, as many scholars suggest (e.g., Cornelissen et al. 2018, Felfe and Lalive 2018), but also through a reduction in inadequate parenting or insufficient informal care arrangements.

Third, our study contributes more broadly to the literature investigating how economic circumstances, public policies, and household composition affect child maltreatment. Starting with Paxson and Waldfogel's (2002) influential work, many studies reveal relationships

⁶ Effects of daycare for children age 3 to 5 years are generally more positive; for older children as well, daycare is most beneficial for more disadvantaged children (e.g. Havnes and Mogstad, 2015, Cornelissen et al. 2018).

between economic hardship, absent fathers, working mothers, and child maltreatment (Berger et al. 2017, Berger and Waldfogel 2011, Raissian and Bullinger 2017, Slack et al. 2003). These studies rely mostly on correlations or have weaknesses in their measures of child maltreatment. However, two recent studies have used small-area time variation and administrative data for identification. For the U.S., Brown and DeCao (2018) find that high unemployment rates increase child neglect. Lindo et al. (2018), using data from California, show that male layoffs increase child maltreatment, while the opposite is true for female layoffs. Our results make an important contribution to these findings by showing that the provision of public childcare creates an opportunity to attenuate the consequences of economic hardship, unemployment and, in particular, inadequate care.

The remainder of the paper is organized as follows. Section 2 provides some theoretical considerations about the relationship between childcare expansion and child maltreatment. Section 3 explains the child welfare system in Germany and our child maltreatment measure. Section 4 describes the public childcare expansion reform in Germany. Section 5 presents the empirical analysis and the identification strategy. The main results are presented in section 6, followed by an analysis of potential mechanisms in section 7. Section 8 discusses the results, policy implications and conclusions.

2. Theoretical Considerations of Potential Mechanisms

The provision of public childcare aims primarily at improving equal opportunities for men and women in the labor market and at offering early childhood education. Although such care does not focus directly on increasing parental skills or reducing child maltreatment, it has the potential to reduce child maltreatment through various channels. To illustrate these channels, we discuss the two most likely reactions of families when more childcare placements become available: switching from home care (where the mother is the main caregiver) to childcare, or from informal care (e.g., where nannies, the father or partner of the mother, or other relatives are the main caregivers) to childcare.⁷

For those families who substitute informal with formal childcare, the quality of care is likely to increase as children will spend less time with potentially inadequate caregivers. This is particularly true for Germany, as the law sets high quality standards for public childcare.⁸ This higher quality may prevent child maltreatment. In line with this explanation, Lindo et al. (2018) demonstrate that cases of maltreatment increase if men, as a main source of informal childcare, spend more time with children. If high-quality formal childcare is available, mothers will be less dependent on potentially inadequate informal care provided by family members and other insufficient care arrangements. The risk of child maltreatment may therefore decrease.

Given the low childcare fees in Germany, an expansion of maternal labor supply with an accompanied switch from home to formal care will likely increase household income for affected families. This possibly reduces parental stress and provides additional resources for the family, thus helping to avoid child maltreatment. Moreover, higher employment may foster certain consistent behaviors, such as a routine daily schedule, and may extend the family's social network. These improved factors may spill over to maternal parenting and have a preventive effect on child maltreatment. However, Paxon and Waldfogel (2002) also discuss potential negative effects of maternal employment, such as possible job stress, more difficulty making ends meet due to work-related expenses, and less energy available for parenting at the day's end.

The expansion of maternal labor supply may also lead to changes in within-household bargaining. Recent research by Aizer (2010), Bobonis et al. (2013), or Anderberg et al. (2015)

⁷ Informal care includes nannies, fathers, grandparents, partners, older siblings, friends and, in the worst case, the child being left alone. Leaving a small child alone is a direct form of neglect. ⁸ The higher quality of formal childcare, compared with informal childcare, is also documented in Datta Gupta and Simonsen (2010), Herbst (2013), and Gathmann and Sass (2018).

shows that increases in female relative to male wages, in public transfers, or in employment opportunities reduce domestic violence against women because of changes in household bargaining power. More available childcare places may also improve the economic situation of women in relation to men and may therefore strengthen the bargaining situation of the mother. This strengthening may lead to a separation from a potential male perpetrator, or a change in the behavior of a potential male perpetrator. Both may not only reduce violence against women but also against children.

Some families switching from home care to childcare may not increase labor supply, as shown in international studies on labor supply elasticities concerning childcare availability (Baker et al. 2008 for the U.S.; Havnes and Mogstad 2015 for Norway; Bauernschuster and Schlotter 2015, Busse and Gathmann 2018, and Müller and Wrohlich 2019 for Germany). In those cases, the provision of universal public care may allow additional leisure time for parents. Research on the origins of child maltreatment shows broad consensus that domestic violence against children is rarely a conscious criminal decision by the parents; instead, parental stress and overburden are frequent starting points, particularly in families with low socioeconomic status, low economic resources, and multiple children.⁹ More parental leisure time may mitigate these burdens.

Finally, formal childcare can give all parents-at-risk the opportunity to interact with nursery staff. This interaction may provide parenting guidance and constitute a substantial source of support. It may reduce overburden and, in turn, prevent maltreatment. Additionally, in allocating places in childcare, CPS may focus particularly on families at risk, be it to relieve

⁹ For example, McLoyd (1990) analyzes the effects of economic hardship on children and shows that "poverty and economic loss diminish the capacity for supportive, consistent, and involved parenting and render parents more vulnerable to the debilitating effects of negative life events" (p. 312). In addition, she notes, "a major mediator of the link between economic hardship and parenting behavior is psychological distress deriving from an excess of negative life events, undesirable chronic conditions, and the absence and disruption of marital bonds" (p. 312). For Germany, Deutsche Kinderhilfe (2014) comes to a similar conclusion.

their stress, to monitor them (which may change the behavior of a potential perpetrator), or to obtain access to the families to connect them with other intensive early childhood interventions. Overall, these measures may reduce the number of child maltreatment cases.

In summary, childcare may improve the quality of care, may give better income prospects to the parents if labor supply increases, may improve the bargaining power of mothers, may reduce overburden, and it may give state authorities the possibility to monitor families at risk or to connect them with other support services.

3. Institutions and Data: Child Maltreatment

Measuring the incidence of child maltreatment cases is challenging because they usually take place in the private domain. The literature has relied on several proxies for child maltreatment. Some studies use self-reports from surveys (e.g., Berger et al. 2017), while more recent studies from the U.S. (e.g., Raissian und Bullinger 2017, Brown and DeCao 2018, Lindo et al. 2018) use administrative data from the CPS. Self-reported data may be subject to reliability problems. For Germany, Sierau et al. (2017), for example, show that parents who are part of a child protection case often do not report maltreatment in the context of psychological questions.

In our study, we use the number of child protection cases as a proxy for child maltreatment. Our data source is the German Child and Youth Welfare Statistics (*Deutsche Kinder- und Jugendhilfe Statistik*), which contains all individual cases of child protection at the county level from 2002 to 2015. According to article 19 of the Convention on the Rights of the Child of the United Nations, "states parties shall take all appropriate legislative, administrative, social and educational measures to protect the child from all forms of physical or mental violence, injury or abuse, neglect or negligent treatment, maltreatment or exploitation, including sexual abuse, while in the case of parent(s), legal guardian(s) or any other person who has the care of the child." In line with the Convention, German law defines a child protection case as the temporary placement of a child with a suitable person or in an adequate location if the child's well-being is in danger.

In Germany, CPS is organized at the county level and is in charge of initiating child protection cases. The local CPS becomes active if the situation in a family – which is often already under the observation of the local CPS – becomes critical for the child. A CPS worker then decides whether a persistent danger to the child's well-being exists. If so, the CPS places the child outside the family.¹⁰ Such cases can end with the child returning to the family either with no obligations or with some obligations, most likely a weekly social worker visit or, in an extreme case, a long-term foster care placement (see Petermann et al. 2014 for details on legal regulation).

For several reasons, our measure of child maltreatment – the number of child protection cases – is a very reliable proxy for the overall incidence of child maltreatment. First, in each reported child protection case, an official authority decided that the well-being of the child was in danger. Therefore, if a child protection case is initiated, serious danger to the child – rather than a potential danger – exists. Second, although an unknown rate of unreported cases remains, for serious cases, this rate is likely to be lower than for less serious cases. For this reason, it is unlikely that the detection rate differs systematically by county. Third, the rules for a child protection case in Germany are defined at the federal level in the German Social Code Book (*Sozialgesetzbuch*). This law defines the precise situations in which the well-being of the child is in danger. Therefore – in contrast to the U.S. – changes in the number of cases and not to changes in definitions of child maltreatment.

¹⁰ A family judge becomes involved only if parents disagree with the initiation of a child protection case.

For each child protection case, the data include the age – categorized into seven age groups – of the protected child.¹¹ In addition to the age of the child, the year, and the county, the data provide the reason for the incidence, the household situation, and the person or institution who reported the case. Data availability is very good: Only one state (Schleswig-Holstein, a small northern state) does not provide data on child protection cases in 2002. In subsequent years, very few counties have missing entries. Overall, our analysis focuses on a final sample of 4,420 county-year cells from 11 German federal states.

< Table 1 about here >

Table 1 gives an overview of the number and type of cases in our analysis sample. Overall, 48,757 child protection cases were initiated for children under age 6 from 2002 to 2015. (For total numbers per year see Appendix I.) The most frequent reasons¹² were overburden of the parents (37%), child neglect (36%) and child abuse (12%). The most frequent household situation in which a maltreated child lived is a single parent household (41%), followed by a nuclear family (33%), a biological parent with a new partner (13%), and other household arrangements (12%). The local CPS reported most of the cases (67%). Parents who seek help by themselves (11%) and police (10%) are the second most frequent reporting sources. Nurseries account for only a small number of reports (1%).

< Figure 1 about here >

The maps in Figure 1 show the development of child protection cases per 1,000 children based on our data over time. The variation in reported incidence rates across counties is considerable. Moreover, we observe an increase in reported cases over time (see Appendix II

¹¹ Age is separated in the following seven age groups: below 3, 3 to 6, 6 to 9, 9 to 12, 12 to 14, 14 to 16, and 16 to 18.

 $^{^{12}}$ The reported reasons are not mutually exclusive and multiple assignments are possible but rarely the case. In the analysis below we impose the following ordering: overburden < neglect < abuse, and use the reason with the highest rank.

for details). The overall increase in reporting may reflect a number of additional causes, including better overall awareness of child well-being¹³ (Witt et al. 2017) and prominent cases of child abuse in Germany (and worldwide) in the mid-2000s that received major media attention.¹⁴ The very low overall report rate from nurseries suggests (see Table 1) that increased reports by nurseries due to higher exposure to public childcare do not appear to be the main cause for the increase in child maltreatment cases.

4. Institutions and Data: The German Childcare Expansion

To identify the effect of public childcare provision on child maltreatment, we use Germany's expansion of childcare places for children under the age of three. This expansion began in 2005, when the German federal government committed to creating 230,000 additional early childcare places in West Germany by 2010.¹⁵ While Germany had introduced laws that mandated the provision of universal public childcare for children between ages three and six in 1996, for children under age three, daycare opportunities in the Western federal states hardly existed until 2005.¹⁶

In 2007, a summit of the federal government, the federal states, and the counties reinforced the aim of the 2005 mandate and set the target of a 35% coverage rate by 2013. Finally, the law

¹³ Corporal punishment of children by parents became illegal in Germany in the year 2000.

¹⁴ For example, in 2010, revelations of abuse scandals in the Roman Catholic Church and in educational institutions triggered a public debate about child maltreatment and generated a range of measures focused on prevention. This debate might have raised awareness about child maltreatment and increased the number of reported cases (e.g., Rassenhofer et al. 2015; Witt et al. 2017).

¹⁵ East Germany experienced a much smaller expansion of public childcare provision during the years under analysis because it already had high childcare levels as a legacy of the former German Democratic Republic. Additionally, numerous changes to the boundaries of East German counties over the years would hamper the empirical analysis. Therefore, our analysis only includes West German counties.

¹⁶ Although laws mandating childcare places for children between 3 and 6 years of age were not introduced until 1996, the provision of daycare spaces was already far higher for this age group at that time. Schmitz et al. (2017) report a coverage rate of 78% in 1994, which increased to 93% in 2016.

on support for children,¹⁷ enacted in December 2008, gave every parent with a child aged one to three the right to a place in early childcare by August 2013 – and, if no place was available, the parent could enforce a legal claim for reimbursement. In essence, the reform included a federal government mandate that all counties in each state had to expand public childcare substantially to meet legal rights to guaranteed childcare places for all preschool children aged one to three by August 2013.

We use administrative data from the Statistical Offices of the German Laender¹⁸ to obtain information on public childcare coverage for children under age three at the county level. These data are available for 2002 and then annually for 2006 to 2015. No administrative data on public childcare provision at the county level are available for the years 2003, 2004 and 2005. Figure 2 shows the county-level coverage rates for all years from 2002 and 2006 to 2015.

While in 2002, the childcare coverage rate was consistently below 5% across virtually all West German counties, in 2015, almost all counties exceeded 20% coverage, with an average of 28% (and therefore still below the target rate of 35%). Nonetheless, the maps further show considerable variation in the expansion across counties – even within the same state. Bauernschuster et al. (2016) note that two-thirds of the variation in childcare coverage is attributable to variation within states, while one-third is attributable to differences between them.

< Figure 2 about here >

A further implication of Figure 2 is that we observe a shift to the right of the entire distribution of childcare coverage. However, no convergence process between counties exists (see also Bauernschuster et al. 2016). Instead, the standard deviation of coverage rates steadily increased from 2002 to 2015 (see Appendix III for details), likely due to different expansion

¹⁷ Kinderförderungsgesetz.

¹⁸ Statistische Landesämter.

patterns across counties (some have expanded very slowly, others very rapidly). Some counties have gradually increased childcare over time, some started off strongly but came to a halt, and still others were delayed by a few years but later drastically increased their coverage. Overall, we observe many different types of expansion patterns across counties, resulting in very strong regional variation.

Bauernschuster et al. (2016) and Felfe and Lalive (2018) explain in detail that this variation resulted from the process of opening up new childcare centers, a process that involves many complex and intertwined decisions by municipality, county, and state authorities. On the one hand, municipality and county authorities were responsible for assessing local demand for childcare, with demographic and economic factors (e.g., current cohort sizes and labor market conditions) entering those projections. On the other hand, state authorities had to approve nonprofit organizations' proposals to set up new childcare centers.

This administrative process was susceptible to problems that varied substantially across counties (e.g., Hüsken 2011). These problems included varying routines and levels of knowledge about the complex (co-)funding system (with subsidies from the federal government, the state, and the municipality), construction land shortages, various building regulations for childcare centers, shortages of qualified childcare workers, serious delays in approval, and rejections of noncompliant applications. As a result, the increase in childcare places differed at the county level due to both well-defined predictors of local childcare demand and shocks to the local supply of new childcare places – shocks resulting from lengthy and intricate administrative processes and rules (e.g., Felfe and Lalive 2018). These shocks, which are arguably orthogonal to expected changes in cases of child maltreatment, provide the basis for our identification strategy.

Childcare centers are subject to strict quality regulations, including for opening hours, group size, staff-child ratios, and staff qualifications. Centers are required to remain open for at least

four hours a day, five days per week. Groups within these centers can have up to 10 children and must be supervised by at least one certified education specialist and one (or two) assistants. The educational degree required for group leaders in a care center requires two years of certified vocational training (in the German apprenticeship system) and at least two years of experience at a care center. During the period under study, the ratio of children to staff was approximately 3 to 1 (Felfe and Lalive 2018).

Childcare for children under age three is highly subsidized in Germany. In 2006, public subsidies covered 79% of total operating costs, with another 7% of funding coming from private organizations. Parents had to bear only 14% of total costs. Parental fees are regressive according to family size and progressive according to family income (means-tested), and they range from 0 to 600 euros per month (Bauernschuster et al. 2016). In almost all communities, childcare is free for families who receive welfare benefits. In addition, these families are on a priority list for receiving a place in childcare. The waiving of fees for welfare-dependent families and the preferred placement allocation they receive are both independent of employment status.

5. Empirical Approach

To identify the effects of universal public childcare provision on child protection cases, we follow the empirical approach applied in studies examining the effects of childcare expansions on child development or fertility (e.g. Berlinski et al. 2009, Havnes and Mogstad 2011, Bauernschuster et al. 2016). This approach estimates a generalized difference-in-differences (DiD) model that uses the local childcare coverage rate as a continuous treatment variable, exploiting the large variation – generated by the expansion – in available childcare places across counties and within counties over time.

This generalized model can be specified as follows:

$$y_{ct} = \theta \ treat_cr_{ct} + X'_{ct}\beta + \alpha_t + \delta_c + \varepsilon_{ct}, \tag{1}$$

where y_{ct} is the logarithm of the number of child protection cases per 1,000 children in county c at time t, and $treat_cr_{ct}$ denotes the childcare coverage rate in county c at time t, i.e., a continuous variable. α_t and δ_c are year- and county-fixed effects.¹⁹ X'_{ct} comprises a set of time-varying county factors that may affect the incidence of child protection cases. Finally, ε_{ct} is the i.i.d. error term. We estimate the model by (weighted) fixed-effects panel regressions. Weights are the county-year population of the analyzed age group. All standard errors are clustered at the county level.

The focus of our analysis is on child protection cases in the age groups 0 to 3 and 3 to 6. Although the childcare expansion focuses on children below three, we choose children under six as our main group of interest because enrollment in childcare is often based on cut-off dates and not on birth dates. One popular cut-off date is September 1st. Children born before a cutoff date attend early childcare until the cut-off date, although they may have already turned three. Therefore, the expansion also affects children older than three, and we consider this by estimating the effects on children in the larger age bracket.

Since we are controlling for county fixed effects, the public childcare expansion need not be orthogonal to time-constant county characteristics. Several other studies have shown that the German childcare expansion is exogenous to time-varying county characteristics (see Felfe and Lalive 2018 and Bauernschuster et al. 2016 for the expansion considered in this study, Cornelissen et al. 2018 for the childcare expansion for children aged three to six years). Nevertheless, we test whether time-varying county characteristics correlate with both the childcare expansion and child protection cases, and we include them as controls in the regressions.

¹⁹ To accommodate for county-year cells with zero observations, we add one case to each cell.

As measures of a county's economic and demographic characteristics in each year, we consider the overall unemployment rate, male and female employment rates, ²⁰ and the share of foreign population. In addition, we consider the annual percentage of school dropouts and the percentage of school graduates with a high school degree (*Abitur*) as proxies of low- and high-skill population rates. Finally, we include the annual share of the population below 6 years and the share of the population between 6 and 18 years, to control for changes in the population's composition related to the childcare expansion.

The identifying assumption is that, conditional on county fixed-effects and the set of timevarying county factors, there are no further unobserved characteristics of a county that vary over time and are correlated with public childcare provision and changes in child maltreatment. We run regressions both with and without the set of county-specific time-varying covariates to investigate the robustness of the estimated effects.

One possible concern is that childcare expansion changes the organizational structure or the available resources of a local CPS. Either could lead to a situation in which the number of child protection cases declines because the child protective service initiates fewer cases, even though the rate of parents who maltreat their children remains the same. To address this concern, we use a placebo group of older children who should also be affected by organizational or resource changes in the CPS, as these offices do not organizationally separate older from younger children. This group, however, should not directly respond to the childcare expansion.

We conduct these placebo estimations using the number of child protection cases for the group of children aged 12 to 18 as the dependent variable in equation 1. With this placebo estimation, we can test whether the CPS changed its procedure in response to the childcare expansion, or whether time variant characteristics change according to the expansion. As the

²⁰ Employment subject to social security contributions, which excludes marginal employment below an income threshold of a monthly salary of 450 Euro.

CPS is responsible for all children under 18, if one of the aspects would be true, we should also see a reduction in this older age group. No reduction, on the contrary, can be interpreted as evidence for the validity of our approach.

Much less of an issue for the question at hand are the responses parents may use to avoid detection, e.g., child perpetrators may move to counties with lower numbers of childcare facilities to avoid protection cases. This behavior is unlikely because parents use childcare voluntarily, and perpetrating parents have easier ways of avoiding detection than moving. In the same sense, we can exclude anticipation effects of childcare expansion with respect to child maltreatment behavior. Anticipation effects would imply a change in child maltreatment behavior before more public childcare places become available, e.g., a parental move made before more childcare spaces become available.

Finally, it is important to note that we interpret our estimation results as lower-bound estimates of the corresponding effects. Because the issue of child maltreatment is subject to a large number of unreported cases (even with a comprehensive administrative framework, such as the one we use), the expansion of childcare places may induce a higher probability of detection in addition to its positive effects on parental behavior. A higher probability of detection in treated counties would therefore reduce the expected positive effect in terms of a lower number of child protection cases due to public childcare provision.

6. Empirical Results

Before turning to the estimation results for our main groups, we test for the exogeneity of the timing of childcare slots' creation across counties. To do so, we regress the childcare coverage rates on sociodemographic characteristics (see Table 2). In the first two columns we use the full sample (with non-missing data in childcare coverage and the controls), while in the last two columns we restrict to county-year cells with non-missing child maltreatment cases.

The first and third columns do not include county or year fixed-effects. Childcare coverage was higher in counties with lower unemployment rates, higher female employment, and with a lower share of children in the population. The second and fourth columns show the results when we include county and year fixed-effects. The covariates are no longer jointly significant. In line with previous research and our argumentation above, these estimates demonstrate that changes in the county sociodemographic characteristics are not significantly correlated with the childcare expansion. Hence, the results in Table 2 support our identifying assumption that both the intensity and timing of new childcare slot creation are plausibly exogenous.²¹

< Table 2 about here >

Table 3 presents our main estimation results for different age groups. Panel A shows the results for children from age group 0 to 6 and for children from age group 12 to 18. Panel B separates children 0 to 6, in a younger group 0 to 3 and the older group of 3 to 6. The coefficients reported can be interpreted as the effect of a one-percentage point increase in childcare coverage on the logarithm of child protection cases per 1,000 children. Column 1 shows the results without time-varying county characteristics, while Column 2 additionally includes socioeconomic characteristics and Column 3 also includes educational and population county characteristics.

< Table 3 about here >

The estimation results in Panel A, Column 1 show that an increase in childcare slots by one percentage point significantly reduces cases of child maltreatment by 1.8%. The results in Columns 2 and 3 are very similar in size and significance, confirming that the childcare expansion was exogenous to our time-varying county characteristics. Columns 4, 5, and 6 of Panel A present the corresponding results for the older children. For this older age group, the

²¹ The female employment rate is potentially a "bad control" since it could be affected by childcare availability. We test for this possibility explicitly in section 7.

coefficients are insignificant and close to zero. The pattern of the estimates is robust, independent of whether we consider time-varying county characteristics in the regression model. The results confirm that childcare expansion had no effect on children and families who were not in the age range affected by the expansion. This finding suggests that the local CPS did not change their behavior in response to the childcare expansion, as it is unlikely that those changes would be limited to small children.

Panel B of Table 3 shows that the results are of similar size for children in the age group 0 to 3 years and in the age group 3 to below 6 years. Again, the results are robust independently of whether we take time-varying county characteristics into consideration or not. The results for both age groups are similar in size. This could be due to some delay in the effects, or to spillover effects between siblings, such that a child below three years attending early childcare may affect the whole family. Because the expansion affects all children below six years, we continue presenting the pooled results for children in the age group 0 to 6.

To investigate the robustness of our main results presented in Table 3, we estimate a number of additional model specifications, shown in Table 4. First, we use the number of cases per 1,000 children as the dependent variable instead of its log (Column 1). The results show that a one-percentage-point increase in childcare significantly reduces child maltreatment by 0.03 cases per 1,000 children, which corresponds to an effect size of 2.7%. We can thus show that the results are not sensitive to counties with zero cases or the distributional form of the dependent variable.

< Table 4 about here >

Second, to maintain the log nature of our outcome of interest but avoid the linearity assumption, we estimate equation 1 with a Poisson model. By doing so, we take account of the count nature of our data and require the specification of the conditional mean only (Wooldridge

1999). Columns 2 shows the results from the Poisson model, which are very close to the main specification.

Third, although in Germany legislation for child protection comes from the federal level, we consider state-year fixed effects as an additional robustness test (Column 3). We do so because legislation changes and institutions at the state level may affect child maltreatment cases and coincide with childcare expansion or utilization of specific groups. For example, some German states introduced targeted programs to prevent child maltreatment in certain years. The coefficient of interest remains essentially unchanged.

Fourth, in addition to the county and year fixed effects, we include linear and quadratic timestate trends. These trend-interactions (Column 4) allow states to follow specific trends and should support the validity that maltreatment trends would be the same in states in the absence of childcare expansion. The main result is robust to the inclusion of state trends.

Finally, specifications (5) and (6) consider the development of treatment effects over time. Column 5 considers the effect of public childcare provision on the log of child protection cases in the following year (lag specification). The corresponding coefficient estimate shows a lasting effect, reducing the number of child protection cases in the following period. Column 6 provides a simple test on potentially reversed causality, i.e., that the number of child protection cases may have forced a stronger expansion of public childcare provision. This does not seem to be the case, as the small and insignificant coefficient estimate implies.

< Figure 3 about here >

We strengthen these results further by estimating a full event-study specification (following Schmidheiny and Siegloch 2020) that includes a set of three leads and three lags for childcare expansion in the county.²² The results are displayed in Figure 3 (specification in levels). The coefficients on the leads are small and statistically indistinguishable from zero. This supports our identifying assumption, as we find no evidence of pre-existing differential trends that are correlated with the treatment. We find large and significant coefficients in years one to three following the expansion, which suggests persistent effects of childcare availability, that appear in our child maltreatment data with a delay of one year.

Overall, the main empirical results are robust to different specifications: A one percentagepoint increase in childcare places leads to a reduction of about 1.8% in cases per 1,000 children. A back-of-the-envelope calculation shows that the effect sizes are meaningful: From 2002 to 2015, in West Germany childcare slots increased by 25 percentage points; this expansion implies that in 2015, 45% fewer cases occurred in comparison to a situation in which the childcare supply had remained at the same level as 2002. Adding up the avoided cases in each year between 2002 and 2015 (see Appendix I for the total number of cases for each year) implies that 20,625 more child maltreatment cases would have occurred if childcare supply had remained at the same level as 2002.

7. Mechanisms

In this section we attempt to identify the core mechanisms by which childcare expansion may reduce child maltreatment cases. As explained in section 2, childcare expansion may affect child maltreatment through various channels. First, it may give better income prospects to parents if their labor supply increases. Second, formal childcare may improve the quality of care as it reduces time in inadequate care. Third, it may improve the bargain position of mothers. Fourth, it may reduce overburden if parents' leisure time increases, and finally, it may give state

²² In order to increase the number of periods included in this specification, we extend our data set with information on childcare slots up to 2017, and impute childcare coverage in 2003 and 2004 using linear interpolation.

authorities the opportunity to monitor families at risk, which may change the behavior of potential perpetrators.

7.1 Descriptive evidence from GSOEP

We first show some descriptive statistics on childcare utilization by household characteristics, to illustrate the types of families more affected by the increase in childcare coverage. Appendix IV shows the fraction of households using formal childcare in 2005, 2010 and 2015, according to the German Socio-Economic Panel (SOEP). The sample includes all households with children under age 3.

Overall, only 9% of households reported that their child attended formal childcare in 2005. This fraction reached almost 34% in 2015. The increase in utilization was more pronounced among single-parent families, reaching almost 40% in 2015 (versus 31.5% for two-parent households).

In 2005, welfare recipients and immigrant households (families with lower socio-economic status on average) were slightly underrepresented in childcare participation. Their utilization, however, increased over time. By 2015, 38% of welfare-receiving households made use of formal daycare, compared with 33% of non-welfare recipients. Utilization was 36% among immigrant households, versus 31% in native families.

The descriptive evidence presented here suggests that lower SES households (single parents, welfare recipients, immigrants) were more affected by the increase in public childcare provision between 2005 and 2015 in Germany.

7.2 Maternal labor force participation

We next analyze in more detail the potential effect of the childcare expansion on female employment rates, to examine the extent to which better income prospects of the parents is a plausible channel for our findings. The regression results are presented in Table 5 (first column). The dependent variable is the female employment rate in a county and year, while the main explanatory variable is the childcare coverage rate. The results suggest that increases in childcare availability led to significantly higher female employment rates in the county. Bauernschuster and Schlotter (2015) and Müller and Wrohlich (2019) also find positive effects of childcare reform on mothers' labor supply.

Figure 4 shows the results from an event-study analysis with three leads and three lags. Our estimates suggest that the effects of the reform on female employment were persistent. Thus, affected families likely increased their labor income, on average. Higher maternal earnings may also increase women's bargaining power within the household. The employment elasticity that we document is however smaller than one, suggesting that not all mothers who accessed childcare expanded their labor supply.

< Table 5 and Figure 4 about here >

7.3 Heterogeneous effects

To examine the remaining potential channels, Table 6 presents the effects of childcare expansion separately for different characteristics (reporting source, family situation, type of maltreatment, and gender) of maltreatment cases. *Panel A* analyses which of the four main *reporting sources* (CPS, police, parents or nursery) reacted most strongly to the childcare expansion. If the CPS reports a maltreatment case that leads to out-of-home placement, the CPS had usually observed the family before this serious maltreatment case occurred. This observation takes place because the families possess risk factors for maltreatment, such as less serious incidences of maltreatment, adverse parenting, or a problematic household structure (e.g. teenage parenthood). Families under observation of the CPS often also show low labor

market attachment; relatedly, 39% of these families received welfare benefits in 2018 (Destatis, 2019).²³

The police report a case if officers can confirm an indication expressed by neighbors or other people. Parents report a case mostly if they feel that they can no longer cope with their situation. If parents or police report the case, it is unlikely that the CPS observed the parents already, or they would have made the report. The last group of reports come from nurseries. Although nurseries do not often report cases, this reporting source may increase in counties where more children attend childcare.

< Table 6 about here >

The estimation results indicate that cases reported by the CPS decrease most strongly (2.6% by a one-percentage-point childcare increase), while we do not observe a significant reduction in any other reporting source. This finding suggests that maltreatment in families at risk, which are already under observation of the CPS, reacts most to the childcare expansion. For these families, childcare appears to stabilize a critical situation, which might have ended in a serious maltreatment case without access to childcare. Finally, nurseries do not report more cases in regions with larger expansions, indicating that in this setting, more childcare utilization does not increase the reporting of maltreatment.

Panel B may help to explain why families at risk – who are often welfare recipients – react most to the childcare expansion. We consider three main *family situations* (single parent, both parents, and single with partner)²⁴ in which maltreatment cases occur. While the childcare expansion has no significant effect on single-parent households, cases of maltreatment decrease

²³ Data on the characteristics and the numbers of families under observation of the CPS are only available for recent years.

²⁴ In Germany, the mother heads more than 90% of all single-parent households with children (BMFSFJ, 2017, p. 12). "Single with partner" means a biological parent (most likely the mother) with a new partner.

significantly in households where an adult male is present, with the strongest decrease in households where the mother lives with a new partner.²⁵ This finding indicates that the presence of a male adult in the household plays an important role in why childcare reduces child maltreatment.

The potential improvement in women's bargaining position (driven by higher earnings) may lead to increased separation from potential male perpetrators. To investigate this channel, we study the effects of the reform on family stability. The second column of Table 5 shows the results of estimating the effect of childcare expansion on divorce rates at the county level. We find a coefficient that is very close to zero and statistically insignificant, and thus conclude that the reform had no detectable effect on marital stability.

A potential explanation why the effect may be concentrated in households with a male adult relates to recent work by Lindo et al. (2018). They show that most of the perpetrators in child maltreatment cases are men. They find that the female-male ratio in abuse cases is 1:4 considering the reduced time that men spend with children, and that male unemployment increases the risk of child maltreatment. These findings lead to a potential explanation for why childcare plays a dominant role in reducing maltreatment cases if a male is present in the household. Because we find the strongest reduction in families already under observation of the CPS, in which the father is often unemployed or irregularly employed, more available childcare strongly reduces the time a child spends with a potential male perpetrator.

From this argumentation, also the answer follows why childcare expansion has only small effects on mostly female-headed single-parent households. In single-parent households, childcare mostly substitutes maternal care. It appears that this substitution has only small effects

²⁵ In their overview on family stability, Brown et al. (2016) show that cohabiting unions are much less stable than marriages, even when children are present. Related to that there is a growing literature showing that family instability has a causal effect on children's development, see, e.g., Lee and McLanahan (2015).

on child maltreatment. An alternative explanation for the small effects for single parents might be lower childcare utilization by single parents than by nuclear families. However, Appendix IV shows that single parents are in fact more likely to use childcare for children below three compared with two parent families, thus allowing us to rule out this explanation.

Panel C of Table 6 presents the estimation results on which *reasons for a maltreatment case* (neglect, abuse, and overburden) are most affected by the childcare expansion. "*Neglect*" indicates that parents are caring for their child insufficiently, whereas "*abuse*" indicates that parents are harming the child. In contrast, "*overburden*" indicates that severe problems of the parents, such as drug addiction or mental illness, which require that the child not remain with the parents, are the main reason for the out-of-home placement. We find a strong decrease in cases initiated because of child abuse and neglect, while we find no significant decrease in cases initiated because of overburden. These findings indicate that the additional free time a family may get due to using childcare does not reduce the severe problems facing the parents. Therefore, the reduction in abuse and neglect as the main driver also points to the direction in which inadequate care is reduced.

Finally, *Panel D* shows that the expansion affects maltreatment cases of *girls* somewhat more than boys. Nevertheless, the effects are significant for both genders.

Our analysis in this section shows that the childcare expansion increased female employment. Thus, higher household income and/or improved female bargaining power appear to be a potential channel behind the reduction in child maltreatment. Additionally, the characteristics of the maltreatment cases suggest that in families at risk, reduced time in inadequate care with a potential male perpetrator may constitute another channel through which childcare reduces maltreatment. Other explanations, such as stronger monitoring by the CPS or more leisure time, appear to contribute to a smaller extent since if these were the main channels, the expansion should also affect single-mother families.

8. Conclusions

We investigate the effects of public childcare provision on child protection cases. For identification, we use an exogenous expansion of childcare places for children below age three in Germany. Our results suggest that the provision of childcare places reduced the number of child protection cases in a meaningful and significant way. We thus show that a large-scale public policy, even one that does not directly aim at preventing child maltreatment, can have beneficial side effects. This finding is both new and important, given that some scholars and child welfare organizations argue that only very intensive and focused interventions can prevent tragic incidences of child maltreatment.

For example, the charity Prevent Child Abuse America argues that home visiting, early childhood education, and parent education are the most effective interventions to prevent child neglect. While also advocating for mental health services for parents, ensuring access for all children to affordable, quality health care and increasing efforts to alleviate social problems such as poverty, Prevent Child Abuse America does not mention public childcare provision as an effective preventive policy.²⁶

Our results further show that maltreatment reduction is strongest in families at risk of maltreatment in which a male adult is present. This finding strengthens the argument for further expanding publicly provided childcare and subsidizing access for low-income groups, who are at the highest risk of child maltreatment. However, because we find no effects for single households, more intensive programs appear to remain important in these households for preventing child maltreatment.²⁷

²⁶ See http://preventchildabuse.org/resource/preventing-child-neglect/ for details.

²⁷ Home visiting programs are the most prominent intensive early childhood intervention for preventing child abuse and neglect. These programs are expanding in both Europe and the U.S. (e.g., U.S. Department of Health and Human Services, 2015, Robling et al. 2016, Sandner et al. 2018).

Finally, our results provide further legitimation for publicly provided childcare, as they show beneficial side effects of this policy. These side effects are fiscally relevant since child maltreatment not only causes extreme hardship for the victims, but also leads to enormous longrun fiscal costs for societies due to increased need for special education, impaired health, and higher welfare payments.

For a back-of-the-envelope cost-benefit analysis, we assume societal costs of 90,000£ (111,600 US-\$) per child maltreatment case. Conti et al. (2019) estimate these costs for the UK, which are lower than the costs calculated by Fang et al. (2012) for the U.S. but probably more comparable to Germany. Because our results show that the childcare expansion prevented about 20,625 cases of serious maltreatment, the expansion generated savings for German society of about \$2.3 billion over the observation period. These savings represent 63% of total federal investments in childcare expansion until 2015, which amounted to \$3.64 billion (3.28 billion Euro). Therefore, preventing child maltreatment adds a substantial amount to other expected positive fiscal benefits of public child care provision, such as tax revenues and social security contributions due to higher female employment and a better educated workforce in the long-run.

References

- Ammerman, R. T., J. E. Cassisi, M. Hersen, and V. B. van Hasselt (1986). Consequences of physical abuse and neglect in children. *Clinical Psychological Review*, 6, 291–310.
- Baker, M., J. Gruber, and K. Milligan (2008). Universal Childcare, Maternal Labor Supply, and Family Well-Being. *Journal of Political Economy*, 116(4), 709–745.
- Baron, E. J., Ezra G. Goldstein, and Cullen T. Wallace (2020). Suffering in silence: How COVID-19 school closures inhibit the reporting of child maltreatment. *Journal of Public Economics* 190.
- Bauernschuster, S., T. Hener, and H. Rainer (2016). Children of a (Policy) Revolution: The Introduction of Universal Childcare and Its Effects on Fertility. *Journal of the European Economic Association*, 14(4), 975–1005.
- Bauernschuster, S. and M. Schlotter (2015). Public Childcare and Mothers' Labor Supply Evidence from Two Quasi-Experiments. *Journal of Public Economics*, 123, 1–16.
- Berger, L. M., S. A. Font, K. S. Slack, and J. Waldfogel (2017). Income and child maltreatment in unmarried families: evidence from the earned income tax credit. *Review of Economics of the Household*, 15(4), 1345–1372.
- Berger, L. M. and J. Waldfogel (2011). Economic determinants and consequences of child maltreatment. OECD Social, Employment and Migration Working Paper No. 111. OECD Publishing, Paris.
- Berlinski, S., S. Galiani, and P. Gertler (2009). The effect of pre-primary education on primary school performance. *Journal of Public Economics*, 93, 219–234.
- Bitler, M., H. Hoynes, and T. Domina (2014). *Experimental Evidence on Distributional Effects* of Head Start. Cambridge, MA: NBER Working Paper No. 20434, August 2014.
- BMFSFJ (2017). Familienreport 2017. German Federal Ministry for Families, Elderly, Women and Youth, Berlin. https://www.bmfsfj.de/bmfsfj/service/publikationen/familienreport-2017/119526.
- Brown, D. and E. DeCao (2018). *The Impact of Unemployment on Child Maltreatment in the United States*. ISER Working Paper Series 2018-04, Institute for Social and Economic Research.
- Brown, S. L., J. B. Stykes, and W. D. Manning (2016). Trends in Children's Family Instability, 1995–2010. *Family Relations*, 78, 1173–1183.
- Busse, A. and C. Gathmann (2018). Free Childcare and Its Effects on Children and Their Families. IZA Working Paper No. 11269, IZA Bonn.
- Carlsson, S., and S. Thomsen (2015). Improving the Allocation of Spots in Child Care Facilities for Toddlers in Germany: A Mechanism Design Approach. IZA DP 8976, Bonn.
- Conti, G., S. Morris, M. Meinychuk, and E. Pizzo (2017). *The economic costs of child maltreatment in the UK: a preliminary study.* London: NSPCC.
- Cornelissen, T., C. Dustmann, A. Raute, and U. Schönberg (2018). Who benefits from universal childcare? Estimating marginal returns to early childcare attendance. *Journal of Political Economy*, 126(6), 2356–2409.

- Cunha, F. and J. J. Heckman (2007). The Technology of Skill Formation. *American Economic Review*, 97(2), 31–47.
- Cunha, F., J. J. Heckman, L. Lochner, and D. Masterov (2006). Interpreting the evidence on life cycle skill formation. In: Hanushek, E.A. and F. Welch (Eds.), *Handbook of the Economics of Education*, Vol. 1, Chr. 12, 697–812. Elsevier, North-Holland, Amsterdam.
- Currie, J. and C. Spatz Widom (2010). Long-term consequences of child abuse and neglect on adult economic well-being. *Child Maltreatment*, 15, 111–120.
- Currie, J. and E. Tekin (2012). Understanding the Cycle Childhood Maltreatment and Future Crime. *Journal of Human Resources*, 47(1), 509–549.
- Datta Gupta, N. and M. Simonsen (2010). Non-cognitive Child Outcomes and Universal High Quality Childcare. *Journal of Public Economics*, 94(1-2), S. 30–43.
- Destatis (2019). 2018 erstmals über 1 Million erzieherische Hilfen für junge Menschen. Destatis Pressemitteilung, Nr. 424 vom 31. Oktober 2019. https://www.destatis.de/DE/Presse/Pressemitteilungen/2019/10/PD19 424 225.html.
- Deutsche Kinderhilfe (2014). Vorstellung der Zahlen kindlicher Gewaltopfer Auswertung der Polizeilichen Kriminalstatistik 2013. Information to the press from April 7. 2014. Accessed: August 13, 2014.
- Doyle, J. J., and A. Aizer (2018) Economics of Child Protection: Maltreatment, Foster Care, and Intimate Partner Violence *Annual Review of Economics*, 10, 87–108.
- Drange, N. and T. Havnes (2019). Early Childcare and Cognitive Development: Evidence from an Assignment Lottery. *Journal of Labor Economics*, 37(2), 581–620.
- Fang. X., D. S. Brown, C. S. Florence, and J. A. Mercy (2012). The economic burden of child mal-treatment in the United States and implications for prevention. *Child Abuse & Neglect*, 36, 156–165.
- Federal Statistical Office (2015). Statistiken der Kinder- und Jugendhilfe: Vorläufige Schutzmaßnahmen 2014. Statistisches Bundesamt, Wiesbaden.
- Felfe, C. and R. Lalive (2018). Does Early Childcare Affect Children's Development? *Journal* of Public Economics, 159, 33–53.
- Fitzpatrick, M, Benson, C and S Bondurant (2020). Beyond Reading, Writing, and Arithmetic: The Role of Teachers and Schools in Reporting Child Maltreatment. NBER Working Paper No. 27033.
- Fort, M., A. Ichino, and G. Zanella (2016). Cognitive and Noncognitive Costs of Day Care at Age 0–2 for Children in Advantaged Families. Forthcoming: *Journal of Political Economy*
- Gathmann, C. and B. Sass (2018). Taxing Childcare: Effects on Childcare Choices, Family Labor Supply and Children. *Journal of Labor Economics*, 36(3), 665–709.
- Heckman, J. J. and S. Mosso (2014). The Economics of Human Development and Social Mobility. *Annual Review of Economics* 6, 689–733.
- Hernández, D. A. Topping, C. Hutchinson, A. Martin, J. Brooks-Gunn, and A. Petitclerc (2019). Client attrition in the Nurse-Family Partnership: Revisiting metrics of impact in a home visitation program in the United States. *Health and Social Care*, 2019; 27:e483–e493.

- Häuser, W., G. Schmutzer, E. Brähler, and H. Glaesmer (2011). Maltreatment in childhood and adolescence—results from a survey of a representative sample of the German population. *Dtsch Arztebl Int*, 108(17), 287–94.
- Havnes, T. and M. Mogstad (2011). No Child Left Behind: Subsidized Childcare and Children's Long-Run Outcomes. *American Economic Journal: Economic Policy*, 3(May 2011), 97–129.
- Havnes, T. and M. Mogstad (2015). Is universal childcare leveling the playing field? *Journal* of *Public Economics*, 127, 100–114.
- Herbst, C. M. (2013). The impact of non-parental childcare on child development: Evidence from the summer participation "dip". *Journal of Public Economics*, 105, 86–105.
- Hildyard, K. L. and D. A. Wolfe (2002). Child neglect: developmental issues and outcomes. *Child Abuse & Neglect*. 26. 679–695.
- Howard, K. and J. Brooks-Gunn (2009). The Role of Home-Visiting Programs in Preventing Child Abuse and Neglect. *The Future of Children*, 19 (2), 119–146.
- Hüsken, K. (2011). Kita vor Ort. Betreuungsatlas auf Ebene der Jugendamtsbezirke 2010. München: DJI.
- Kottelenberg, M. J. and S. F. Lehrer (2017). Targeted or Universal Coverage? Assessing Heterogeneity in the Effects of Universal Childcare. *Journal of Labor Economics* 35(3), 609–653.
- Lee, D. and S. McLanahan (2015). Family Structure Transitions and Child Development: Instability, Selection, and Population Heterogeneity. *American Sociological Review*, 80(4), 738–763.
- Levey E. J., B. Gelaye, P. Bain, M. B. Rondon, and C. P. Borba et al. (2017). A systematic review of randomized controlled trials of interventions designed to decrease child abuse in high-risk families. *Child Abuse Neglect*, 65, 48–57.
- Lindo, J., J. Schaller, and B. Hansen (2018). Caution! Men Not at Work: Gender-Specific Labor Market Conditions and Child Maltreatment. *Journal of Public Economics*, 163, 77–98.
- Magnuson, K. A., C. Ruhm, and J. Waldfogel (2007). Does Prekindergarten improve school preparation and performance?, *Economics of Education Review*, 26(1), 33–51.
- McLoyd, V. (1990). The Impact of Economic Hardship on Black Families and Children: Psychological Distress, Parenting, and Socioemotional Development. *Child Development*, 61(2), 311–346.
- Müller, K.-U. and K. Wrohlich (2019). *Does subsidized care for toddlers increase maternal labor supply? Evidence from a large-scale expansion of early childcare*, CEPA DP N. 9.
- Paxson, C. and J. Waldfogel (2002). Work, Welfare, and Child Maltreatment. *Journal of Labor Economics*, 20(3), 435–475.
- Peter, F., P. S. Schober, and C. K. Spieß (2016). Early Birds in Day Care: The Social Gradient in Starting Day Care and Children's Non-cognitive Skills. *CESifo Economic Studies*, 62(4), 725–751.
- Petermann, F., T. Besier, P. Büttner, S. Rücker, M. Schmid, and J. M. Fegert (2014). Vorläufige Schutzmaßnahmen für gefährdete Kinder und Jugendliche Inobhutnahmen in Deutschland. *Kindheit und Entwicklung*, 23(2), 124–133.

- Rassenhofer M., A. Zimmer, N. Sprober, and J. M. Fegert (2015). Child sexual abuse in the Roman Catholic Church in Germany: comparison of victim-impact data collected through church-sponsored and government-sponsored programs. *Child Abuse and Neglect*, 40, 60–67.
- Raissian, K. M. and L. R. Bullinger (2017). Money matters. Does the minimum wage affect child maltreatment rates? *Children and Youth Services Review*, 72, 60–70.
- Robling, M., M.-J. Bekkers, K. Bell, C. C. Butler, R. Cannings-John, and S. Channon et al. (2016). Effectiveness of a Nurse-Led Intensive Home-Visitation Programme for First-Time Teenage Mothers (Building Blocks): A Pragmatic Randomised Controlled Trial. *The Lancet*, 387 (10014), 146–155.
- Sandner, M. (2019). Effects of early childhood intervention on maternal employment, fertility and well-being Evidence from a randomized controlled trial. *Journal of Health Economics*, Vol. 63, 159–181.
- Sandner, M., T. Cornelissen, T. Jungmann, and P. Herrmann (2018). Evaluating the effects of a targeted home visiting program on maternal and child health outcomes. *Journal of Health Economics* 58, 269–283.
- Schmidheiny, K. and Siegloch, S. (2020). On event studies and distributed-lags in two-way fixed effects models: Identification, equivalence, and generalization. ZEW Discussion Papers 20-017, ZEW - Centre for European Economic Research, Mannheim.
- Schmitz, S., C. K. Spieß, and J. Stahl (2017). Kindertageseinrichtungen: Ausgaben der Familien sind von 1996 bis 2015 mitunter deutlich gestiegen. *DIW Wochenbericht*, 41/2017, 889– 903.
- Sierau, S., T. Brand, J. T. Manly, A. Schlesier-Michel, A. M. Klein, A. Andreas, L. Quintero Garzón, J. Keil, M. J. Binser, K. von Klitzing, and L. O. White (2017). A Multisource Approach to Assessing Child Maltreatment from Records, Caregivers, and Children. *Child Maltreatment*, 22(1), 45–57.
- Slack, K. S., J. L. Holl, B. J. Lee, M. McDaniel, L. Altenbernd, and A. B. Stevens (2003). Child protective intervention in the context of welfare reform: The effects of work and welfare on maltreatment reports. *Journal of Policy Analysis and Management*, 22(4), 517–536.
- Springer, K., J. Sheridan, D. Kuo, and M. Carnes (2007). Long-term Physical and Mental Health consequences of Childhood Physical Abuse: Results from a Large Population-based Sample of Men and Women. *Child Abuse & Neglect*, 31(5), 517–530.
- Statistisches Bundesamt (2016). Statistiken der Kinder- und Jugendhilfe Gefährdungseinschätzungen nach § 8a Absatz 1 SGB VIII. Statistisches Bundesamt, Wiesbaden 2016.
- Thiel, H. and S. L. Thomsen (2013). Noncognitive skills in economics: Models, measurement, and empirical evidence. *Research in Economics*, 67(2), 189–214.
- U.S. Department of Health & Human Services, Administration for Children and Families, Administration on Children, Youth and Families, Children's Bureau (2017). *Child Maltreatment 2015*. Available from <u>http://www.acf.hhs.gov/programs/cb/research-data-technology/statistics-research/child-maltreatment</u>.

- U.S. Department of Health and Human Services (2015). *HHS Awards \$386 Million to Support Families Through the Home Visiting Program:* http://www.hhs.gov/news/press/2015pres/02/20150219a.html.
- U.S. Department of Health & Human Services (DHHS) (2014). *Head Start Services*, available at <u>http://www.acf.hhs.gov/programs/ohs/about/head-start</u>.
- Wang, C. T. and J. Holton (2007). *Total Estimated Cost of Child Abuse and Neglect in the United States*. Prevent Child Abuse America. Chicago. Illinois.
- Witt, A., R. C. Brown, P. L. Plener, E. Brähler, and J. M. Fegert (2017). Child Maltreatment in Germany: Prevalence Rates in the General Population. *Child and Adolescent Psychiatry and Mental Health*, 11, 47.
- Wooldridge, J. M. (1999). Distribution-free estimation of some nonlinear panel data models. *Journal of Econometrics*, 90 (1), 77–97.



Figure 1: Child protection cases per 1,000 children by county in West Germany between 2002 and 2015.

Avg. cases of child protection cases by child welfare authorities in Germany, cases per 1,000 children

Notes: Data provided by the German Child and Youth Welfare Statistic on individual cases of child protection in a particular year on the county level of children under six years of age (per 1,000 children). Numbers in brackets refer to number of counties in each class in 2015. Calculations by the authors.



Figure 2: Childcare expansion by county in West Germany between 2002 and 2015.

Notes: Data provided by the Statistical Offices of the German Laender on public childcare coverage for children under the age of three. Numbers in brackets refer to number of counties in each class in 2015. Calculations by the authors.

30 -< 40 (93) **3** > 40

(8)



Figure 3. The effect of childcare expansion on child maltreatment: Event-study results

Note: The figure displays coefficients and their 95% confidence intervals from a regression where the dependent variable is the number of child maltreatment cases per 1,000 children in a county, and the main explanatory variable is the year-to-year change in the childcare coverage rate, as well as its lags and leads (the first lead is omitted). We also control for year and county fixed-effects. We use data on child maltreatment for years 2002 to 2015, and on childcare coverage for years 2002 to 2017. All observations are weighted by the county population of the age group (0-6). Data are provided by the German child and youth welfare statistic. Standard errors are clustered at the county level.





Note: The figure displays coefficients and their 95% confidence intervals from a regression where the dependent variable is the female employment rate in a county, and the main explanatory variable is the year-to-year change in the childcare coverage rate, as well as its lags and leads (the first lead is omitted). We also control for year and county fixed-effects. We use data on female employment for years 2002 to 2015, and on childcare coverage for years 2002 to 2017. All observations are weighted by the county population of the age group (0-6). Standard errors are clustered at the county level.

Gender MaleMeanGender Male0.54Reason for the Case0.54Reason for the Case0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11		Age: 0 to <6
Gender Male0.54Reason for the Case Overburden of Parents0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the Case Single Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11		Mean
Male0.54Reason for the Case0Overburden of Parents0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by0Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Gender	
Reason for the Case0.37Overburden of Parents0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested byVouth OfficeYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Male	0.54
Reason for the CaseOverburden of Parents0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested byVouth OfficeYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11		
Overburden of Parents0.37Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested byVouth OfficeYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Reason for the Case	
Neglect0.36Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Overburden of Parents	0.37
Abuse and Sexual Abuse0.12Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Neglect	0.36
Other0.26Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested byVouth OfficeYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Abuse and Sexual Abuse	0.12
Living Arrangement Before the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested byYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Other	0.26
Living Arrangement Defore the CaseSingle Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by0.67Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Living Arrangement Pefore the Case	
Single Parent0.41Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Single Deport	0.41
Both Parents0.33Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Single Parent	0.41
Single Parent with Partner0.13Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Both Parents	0.33
Relatives, Foster Family, Other0.12Case was Suggested by Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Single Parent with Partner	0.13
Case was Suggested byYouth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Relatives, Foster Family, Other	0.12
Youth Office0.67Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Case was Suggested by	
Parents0.11Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Youth Office	0.67
Police0.10Nursery/Teacher0.01Medical System, Relatives, Other0.11	Parents	0.11
Nursery/Teacher0.01Medical System, Relatives, Other0.11	Police	0.10
Medical System, Relatives, Other 0.11	Nursery/Teacher	0.01
N 1 60	Medical System, Relatives, Other	0.11
N 1 60 40 555		
Number of Cases 48,757	Number of Cases	48,757

Table 1: Sample descriptive statistics

Notes: Data provided by the German child and youth welfare statistic for the years 2002 to 2015.

Table 2. Balancing test

Unemployment rate	-0.488*** (0.158)	-0.231** (0.110)	-0.484*** (0.158)	-0.209* (0.112)
Share of women in the workforce	0.688***	0.191	0.699***	0.0988
Share of men in the workforce	(0.106)	(0.167)	(0.106) 0.0156	(0.164)
	(0.0796)	(0.119)	(0.0791)	(0.118)
Share of foreign population	-0.152**	-0.0355	-0.176**	-0.0313
Share of the population below 6	(0.0745)	-0.756	(0.0751)	(0.113) -0.989*
	(0.600)	(0.537)	(0.613)	(0.539)
Share of the population between 6 and 18	-3.538***	-0.108	-3.568***	-0.179
	(0.333)	(0.375)	(0.335)	(0.380)
Share of school drop-outs (w/o degree)	-26.13** (11.47)	-0.371 (1.645)	-25.21** (11.36)	-0.537 (1.594)
Share of school graduates with high- school degree	3.109***	0.0691	2.998***	0.0887
Ū	(1.131)	(0.223)	(1.120)	(0.220)
County fixed-effects	no	yes	no	yes
Year fixed-effects <i>p</i> -value for joint significance of	no	yes	no	yes
covariates	0.000	0.1991	0.000	0.2144
Observations	3,511	3,511	3,444	3,444

Notes: The table reports coefficients from regressions of the childcare rate on county characteristics. Each column reports the results from a different regression, where the dependent variable is always the childcare rate. We report the *p*-value for the hypothesis that the county characteristics are jointly equal to zero. All coefficients on shares refer to 1-percentage-point changes in these shares. The last 3 columns drop county-year observations with 0 or missing cases of child maltreatment. Standard errors clustered at the county level are reported in parentheses. Statistical significance indicated by stars (* p<0.1, ** p<0.05, *** p<0.01).

Panel A	Age 0 to <6 years Log cases per 1,000 children			Ag Log ca	ars hildren	
	(1)	(2)	(3)	(4)	(5)	(6)
Effects of one pp. childcare expansion	-0.018**	-0.017**	-0.017**	0.002	0.001	0.002
	(0.008)	(0.008)	(0.008)	(0.006)	(0.006)	(0.006)
Year fixed effects	yes	yes	yes	yes	yes	yes
County fixed effects	yes	yes	yes	yes	yes	yes
Regional economic factors	no	yes	yes	no	yes	yes
Regional educational factors	no	no	yes	no	no	yes
County year observations	3,496	3,469	3,447	3,496	3,469	3,447
Number of counties		324			324	

Table 3: Effects of universal public childcare provision on child protection cases (log cases per 1,000 children).

hel B Age 0 to <3 years Log cases per 1,000 children		Age 3 to <6 years Log cases per 1,000 childr		rs hildren		
	(1)	(2)	(3)	(4)	(5)	(6)
Effects of one pp. childcare expansion	-0.018*	-0.017*	-0.016*	-0.020***	-0.019***	-0.019***
	(0.009)	(0.008)	(0.011)	(0.007)	(0.007)	(0.007)
Year fixed effects	yes	yes	yes	yes	yes	yes
County fixed effects	yes	yes	yes	yes	yes	yes
Regional economic factors	no	yes	yes	no	yes	yes
Regional educational factors	no	no	yes	no	no	yes
County year observations	3,496	3,469	3,447	3,496	3,469	3,447
Number of counties		324			324	

Notes: The table reports coefficients from regressions of the log of child maltreatment cases per 1,000 children on childcare rate for years 2002 to 2015. Regional economic and educational factors include the variables shown in Table 2. All observations are weighted by the county population of the observed age group. Data are provided by the German child and youth welfare statistic. Standard errors reported in parentheses are clustered on the county level. Statistical significance indicated by stars (* p<0.1, ** p<0.05, *** p<0.01).

	Age 0 to <6 years					
	Cases per 1,000 children (1)	Cases per 1,000 children (Poisson regression) (2)	Log cases per 1,000 children (3)	Log cases per 1,000 children (4)	Log cases per 1,000 children (childcare t- 1) (5)	Log cases per 1,000 children (childcare t+1) (6)
Effects of one pp. childcare	-0.030***	-0.018*	-0.018**	-0.020**	-0.019***	-0.003
expansion	(0.010)	(0.010)	(0.009)	(0.009)	(0.007)	(0.009)
Year fixed effects	yes	yes	yes	yes	yes	yes
County fixed effects	yes	yes	yes	yes	yes	yes
State-year fixed effects	no	no	yes	no	no	no
Linear and quadratic state trends	no	no	no	yes	no	no
County year observations			3,496	ī		
Number of counties			324			

Table 4: Effects of universal public childcare provision on child protection cases - different model specifications.

Notes: The table reports coefficients from regressions of different specifications of equation (1). All observations are weighted by the county population of the observed age group. Columns 1-4 show the effect of childcare expansion in t while column 5 shows the effect of childcare expansion in t-1 and column 6 in t+1. Standard errors reported in parentheses are clustered on county level. Statistical significance indicated by stars (* p<0.1, ** p<0.05, *** p<0.01).

 Table 5. Effects of public childcare provision on female employment and divorce

	Female employment rate	Divorce rate
Effects of 1pp. childcare expansion	0.0566*** (0.0171)	0.0022 (0.0040)
Year fixed effects	yes	yes
County fixed effects	yes	yes
County-year observations	4,536	3,052
Number of counties	324	218

Notes: The table reports coefficients from regressions of equation (1), the female employment rate or the divorce rate in the county as the dependent variable. Estimations are based on fixed-effects panel regressions for years 2002 to 2015. All observations are weighted by county population age 0 to 6. Standard errors reported in parentheses are clustered at the county level. Statistical significance is indicated by stars (* p<0.1, ** p<0.05, *** p<0.01).

Table 6: Effect for subgroups for age 0 to <6</th>

	Youth		_	
Panel A: Who reported the Case	Office	Police	Parents	Nursery
Effects of one pp. childcare expansion	-0.026**	-0.001	-0.013	-0.000
	(0.010)	(0.007)	(0.008)	(0.004)
		Both	Single with	
Panel B: Living Arrangement Before Case	Single	parents	partner	
Effects of one pp. childcare expansion	-0.009	-0.017**	-0.030***	
	(0.008)	(0.005)	(0.002)	
Panel C: Reason for the Case	Overburden	Neglect	Abuse	
Effects of one pp. childcare expansion	-0.006	-0.017**	-0.019***	
	(0.010)	(0.008)	(0.006)	
	Deer			
Panel D: Gender	Воу	Girl		
Effects of one pp. childcare expansion	-0.015**	-0.023***		
	(0.007)	(0.009)		
Year fixed effects	yes	yes	yes	yes
County fixed effects	yes	yes	yes	yes

Notes: The table reports coefficients from regressions of equation (1) with log cases per 1,000 children in the specific category as dependent variable. Estimations are based on fixed-effects panel regressions for years 2002 to 2015. All observations are weighted by county population age 0 to 6. Standard errors reported in parentheses are clustered on the county level and are robust. Statistical significance indicated by stars (* p<0.1, ** p<0.05, *** p<0.01).

Year	Total number of cases, age < 6	
2002	2,150	
2003	2,147	
2004	2,085	
2005	2,149	
2006	2,566	
2007	3,050	
2008	3,823	
2009	3,865	
2010	4,044	
2011	4,277	
2012	4,604	
2013	4,523	
2014	4,760	
2015	5,116	
All years	49,160	

Appendix I: Total cases of child protection age < 6 in each year

Notes: Data provided by the German Child and Youth Welfare Statistic on absolute individual cases of child protection in a particular year on the county level of children under six years of age. Calculations by the authors. Cases of counties with missing data are imputed by the average cases per county within the state and the specific year.

	Average cases per 1000 children				
		Age	: < 6	Age: betweer	n 12 and < 18
Year	Counties	Mean	S.D.	Mean	S.D.
2002	286	0.501	0.753	1.873	2.365
2003	296	0.501	0.657	1.815	2.331
2004	311	0.534	0.672	1.832	2.377
2005	317	0.547	0.691	1.893	2.441
2006	316	0.690	0.877	1.927	2.400
2007	319	0.872	1.122	2.096	3.091
2008	322	1.089	1.179	2.437	2.717
2009	322	1.123	1.077	2.554	3.076
2010	319	1.198	1.223	2.707	2.866
2011	321	1.319	1.348	2.813	2.837
2012	321	1.252	1.191	2.970	3.078
2013	323	1.289	1.223	2.957	2.932
2014	323	1.317	1.261	3.063	3.307
2015	324	1.361	1.394	2.930	2.774
All years	4,420	0.980	1.132	2.430	2.817

Appendix II: Average cases of child protection age < 6 and between 12 and < 18.

Notes: Data provided by the German Child and Youth Welfare Statistic on absolute individual cases of child protection in a particular year on the county level of children under six years of age. Calculations by the authors.

Year	Counties	Mean	S.D.	Min	Max
2002	324	0.022	0.023	0.000	0.130
2006	324	0.073	0.038	0.010	0.233
2007	324	0.094	0.044	0.022	0.284
2008	324	0.118	0.047	0.034	0.340
2009	324	0.142	0.049	0.036	0.344
2010	324	0.171	0.053	0.071	0.360
2011	324	0.200	0.060	0.092	0.378
2012	324	0.222	0.059	0.110	0.392
2013	324	0.242	0.060	0.113	0.432
2014	324	0.270	0.058	0.139	0.469
2015	324	0.274	0.058	0.136	0.472

Appendix III: Childcare coverage over time.

Note: Data provided by the German Child and Youth Welfare Statistic. The figures show mean childcare coverage rates across West German counties as well as standard deviations, median, minimum, and maximum values. All information is provided for the years 2002 and 2006 to 2015. Calculations by the authors.

	2005	2010	2015
All households	8.9	26.9	33.6
Two-parent households	8.5	22.5	31.5
Single-parent households	10.9	18.6	39.6
Households receiving welfare	8.0	15.0	37.8
Households not on welfare	9.0	28.6	33.3
Immigrant households	6.0	25.0	35.6
Non-immigrant households	10.2	28.3	31.0
Number of households	1,003	4,329	2,079

Appendix IV: Childcare utilization by family characteristics

Notes: The results come from the German Socio-Economic Panel Study (SOEP) for the years 2005, 2010, and 2015. The sample includes all households with children below 3 years of age. Figures in percent (except numbers of households).